Author's response to reviews

Title: Nutritional Rehabilitation in Anorexia Nervosa: Implications for Treatment

Authors:

Enrica Marzola (enrica.marzola@gmail.com)
Jennifer A Nasser (jnasser@att.net)
SA Hashim (phash17@yahoo.com)
Pei-an Betty Shih (pbshih@ucsd.edu)
Walter H Kaye (wkaye@ucsd.edu)

Version: 2 Date: 28 April 2013

Author's response to reviews: see over
April 22, 2013

Mr. Carlo Rye Chua
BMC Psychiatry.

Dear Carlo Rye Chua,

I have recently received your request about permissions to reproduce the figures of the manuscript “Nutritional Rehabilitation in Anorexia Nervosa: Implications for Treatment” MS: 2083452275961654. Please find attached the revised figure legends.

As requested, a point-by-point response to your concerns has been included in this cover letter:
- figure 1, 2 and 4: we changed the legend as follows: “Kaye et al., unpublished data”;
- figure 3: it is a figure I used in a previous paper and we need to obtain permission from the American Journal of Clinical Nutrition journal; please let me know if you can handle this process or if you prefer me to do it;
- figure 5: we specified “original figure, no permission required”.

Please let me know if you have any questions or need further information.

Sincerely,

[Signature]

Walter Kaye, MD
Professor of Psychiatry
Director, UCSD Eating Disorder Research and Treatment Program
UCSD Department of Psychiatry
Chancellor Park
4510 Executive Dr., Suite 315
San Diego, CA 92121
http://eatingdisorders.ucsd.edu
wkaye@ucsd.edu
whkaye@gmail.com
858 205 7293 cell
858 534-3951 office
858 534 6727 fax